Author's response to reviews

Title: Gene expression of sternohyoid and diaphragm muscles in type 2 diabetic rats

Authors:

Erik van Lunteren (exv4@cwru.edu)
Michelle Moyer (michelle.moyer@va.gov)

Version: 2 Date: 16 May 2013

Author's response to reviews: see over
We would like to re-submit the manuscript entitled: "Gene expression of sternohyoid and diaphragm muscles in type 2 diabetic rats" to be considered for publication.

We have requested temporary for reviewers to the GEO data and have now included the link in the manuscript:


We have re-formatted the manuscript so that it conforms to the journal style, including the addition of Competing Interests and Authors’ Contributions sections.